Author's response to reviews

Title: Efficacy of tranexamic acid in reducing blood loss in posterior lumbar spine surgery for degenerative spinal stenosis with instability: a retrospective case control study

Authors:

Stefan Endres (s.endres@elisabeth-klinik.de)
Martin Heinz (simonmartinheinz@gmx.de)
Axel Wilke (a.wilke@elisabeth-klinik.de)

Version: 7 Date: 28 October 2011

Author's response to reviews:

The required changes in author list, removing line numbers and keywords as well as author contribution section was done.